Author's response to reviews

Title: Conjunctival Lymphoma during pregnancy: A Case Report

Authors:
Sherine Jue Ong (sherina8350@yahoo.com.tw)
Shih-Ming Jung (ming22@cloud.cgmh.org.tw)
Hsin-Chiung Lin (lin.a0958@gmail.com)

Version: 1 Date: 16 Jun 2017

Author’s response to reviews:

2017/6/9

Dear Dr. Awadein,

Thank you for your kind review of our work and for the valuable reviewer comments. We have read the comments carefully and revised the manuscript based on their input. Below we have included the comments and our response to them. Changes to the manuscript have been highlighted in yellow.

We are very grateful for the opportunity to submit a revised manuscript, and are looking forward to your next correspondence.

Your Sincerely,

Hsin-Chiung Lin, MD

Response to reviewer comments

Dr. Sava Barisic, PhD (Reviewer 1): Well written case-report of rather uncommon presentation of conjunctival lymphoma.

I would recommend only a few language / style corrections:

1. Page 4 / Line 58: Try rephrasing the sentence ending "there was no new development of her health", with something like "there was no change in her health" or "there was no influence on her health".
2. Page 5 / Line 20: Change word order from "Diagnosis often is made on..." to "Diagnosis is often made on..."

3. Page 5 / Line 36 and 42: Move superscript numbers for in-text citation (12 and 13) to the end of the sentence. It is the style you've been using throughout the rest of your paper.

There is no need for control group, since this is a case report.

Author response:

Thank you for your comments. We have re-write the sentences in page 4 -5 and revised the references in page 5 accordingly.

Dr. Ceyhun Arici (Reviewer 2): The authors studied a case about conjunctival lymphoma during pregnancy. Different case series and clinical studies are present about conjunctival lymphoma.

The study can make contribution on literature by showing the effect of pregnancy on conjunctival lymphoma.

Author response:

Thank you for your comments.